

PHYSICAL STRAIN OF WALKING IN PEOPLE WITH NEUROMUSCULAR DISEASES IS HIGH AND RELATES TO STEP ACTIVITY IN DAILY LIFE

Sander OORSCHOT, MSc^{1,2}, Eric L. VOORN, PhD^{1,2}, Annerieke C. VAN GROENESTIJN, PhD^{1,2}, Frans NOLLET, PhD^{1,2} and Merel A. BREHM, PhD^{1,2}

From the ¹Department of Rehabilitation Medicine, Amsterdam UMC location University of Amsterdam, Amsterdam, The Netherlands; ²*Amsterdam Movement Sciences, Rehabilitation & Development, Amsterdam, The Netherlands*

Objective: To determine the physical strain of walking and assess its relationship with daily steps and intensity of daily activity in people with neuro-muscular diseases.

Design: Cross-sectional study.

Subjects/patients: Sixty-one adults with neuromuscular diseases.

Methods: Physical strain of walking, defined as oxygen consumption during comfortable walking relative to peak oxygen uptake. Daily step count and daily time spent in moderate and vigorous physical activity were assessed using accelerometry and heart rate measurements, respectively. Regression analyses assessed the relationships between log daily step count and log daily time spent in moderate and vigorous physical activity, and physical strain of walking.

Results: The mean (standard deviation) physical strain of walking was 73 (20)% Log daily step count and physical strain were negatively associated ($\beta = -0.47$). No association was found with log daily time spent in moderate and vigorous physical activity.

Conclusions: The highly increased physical strain of comfortable walking indicates that walking is very demanding for people with neuromuscular diseases and is associated with a reduction in daily step activity. The absence of a relationship between intensity of activities and physical strain indicates that, despite a reduction in daily step activity, strenuous daily activities may still be performed.

Key words: neuromuscular disorders; activities of daily living; physical exertion; exercise test; physical therapy.

Submitted Feb 9, 2024. Accepted after revision May 13, 2024

Published Jun 7, 2024. DOI: 10.2340/jrm.v56.40026

J Rehabil Med 2024; 56: jrm40026.

Correspondence address: Sander Oorschot, Department of Rehabilitation Medicine, Amsterdam UMC, location AMC, Meibergdreef 9, Amsterdam, 1105 AZ, the Netherlands. E-mail: s.oorschot@amsterdamumc.nl

Physical activity positively affects physical health, mental health, and quality of life and can reduce the risk of various chronic diseases (1). In people with neuromuscular diseases (NMD), physical activity is reduced in comparison with the general population

LAY ABSTRACT

The percentage at which comfortable walking is performed relative to a person's exercise capacity is called the physical strain of walking. An increased physical strain of walking might reduce daily activity to preserve energy for other tasks. In this study, the physical strain of walking was determined in 61 adults with neuromuscular diseases, and we assessed the relationships between physical strain of walking and daily activity. We found that the physical strain of comfortable walking was highly increased, and walking is thus very demanding for people with neuromuscular diseases. Also, people with high physical strain took fewer steps. Understanding the profiles of daily activity and associated factors can guide rehabilitation interventions to address daily physical functioning. Future studies should investigate whether the physical strain of walking can be reduced by increasing the exercise capacity or reducing the energy required during walking, and analyse whether this improves daily activity.

(2, 3), due to muscle weakness (4), fatigue (5), and pain (6). Furthermore, disease severity (7), walking limitations (4), fear of falling (8), lower self-efficacy (9), and higher age and BMI (8) are reportedly associated with reduced physical activity in NMD. Yet, as shown in polio survivors, these factors, or combinations thereof, explained only 14–31% of the variance in physical activity (4, 8), suggesting the importance of other factors.

Based on findings in cerebral palsy (10), an elevated physical strain of walking might also be associated with reduced physical activity in NMD. Physical strain of walking is defined as the oxygen consumption of walking (VO_{2walk}) relative to the peak oxygen uptake (VO_{2peak}). In people with NMD, the VO_{2peak} has been shown to be up to 50% lower, compared with age- and sex-matched healthy individuals (11, 12), while the oxygen consumption of walking (VO_{2walk}) is reportedly higher (13) due to deviations in the gait pattern (14, 15). A reduced VO_{2peak} together with an increased VO_{2walk} results in an elevated physical strain of walking. In studies among polio survivors, the physical strain of walking (16) and submaximal cycling (17) was found

JRM

to be 65% and 75%, respectively, which is much higher than the physical strain of walking of 34-49% in able-bodied individuals (18, 19). Yet, the studies in polio survivors (16) used predicted maximal heart rate values instead of actual measured VO_{2peak} values, which is less accurate.

A high physical strain of walking may negatively impact daily physical activity (10). Physical activity is often measured by daily step count, which has been shown to be 30% lower in NMD compared with healthy individuals (20). The intensity of physical activity can be categorized according to the time spent in different intensity zones, whereby moderate to vigorous physical activity (MVPA) is recommended by health authorities (21) to reduce the risk of morbidity and mortality (22). Studies in NMD on the time spent in intensity zones reported inconsistent results. Studies mainly using questionnaires (7, 23) reported that people with NMD met the physical activity recommendations, whereas other studies, mostly using accelerometry, gave the opposite results (24-26). This inconsistency in results emphasizes the need for more studies objectively determining the time spent in intensity zones. Moreover, the physical strain of walking in NMD and the associations with both the quantity and intensity of daily activity are unknown, warranting study thereof in this population.

We aimed (*i*) to determine the physical strain of walking based on actual measured VO_{2walk} and VO_{2peak} in individuals with NMD, and (*ii*) to assess the relationship between daily physical activity, expressed as daily step count and time spent in MVPA, and physical strain of walking.

METHODS

Data for this cross-sectional study were obtained from a randomized controlled trial, I'M FINE, investigating the efficacy of a physical activity programme for people with slowly progressive NMD (27). For this study, we used the data collected at baseline. The I'M FINE study was approved by the medical ethics committee of Amsterdam UMC, location Academic Medical Center (AMC), the Netherlands, and registered at the Dutch Trial register (NL7344). All subjects signed informed consent before study enrolment. Study reporting is in accordance with the 'STrengthening the Reporting of OBservational studies in Epidemiology' (STROBE) guidelines.

Participants

Participants in the I'M FINE study were recruited between September 2018 and April 2022 from the departments of Rehabilitation Medicine of two university hospitals and four rehabilitation centres throughout the Netherlands, and through the nationwide Dutch patient organization for NMD (Spierziekten Nederland). Potentially eligible individuals received an information letter. After a phone call to assess general inclusion criteria, individuals willing to participate were invited for a screening visit by a rehabilitation physician. Main inclusion criteria were: diagnosed with Charcot-Marie-Tooth disease, post-polio syndrome, or other slowly progressive NMD; aged \geq 18 years; and motivated to follow a physical activity programme to improve aerobic capacity. Participants were excluded if they had contraindications for physical activity according to the American College of Sports Medicine (ACSM) guidelines (28); were unable to follow verbal or written instructions; had insufficient understanding of the Dutch language; or had participated in an exercise programme for more than 4 weeks in the past 6 months (27). Only ambulatory participants with complete and valid baseline data for all variables of interest, as defined below, were included in the analyses of the current study.

Measurements

Measurements were performed at the Department of Rehabilitation Medicine of the AMC. First, participant characteristics and demographics were registered. Furthermore, muscle strength of 8 muscle groups of both legs (hip flexion/extension, hip abduction/adduction, knee flexion/extension, ankle plantar/ dorsal flexion) was determined by manual muscle testing, and scored according to the Medical Research Council (MRC) scale (29). Scores of all muscle groups were summed to calculate a composite muscle strength score, referred to as MRC sum score (range 0–80). Subsequently, participants performed a walking energy cost test (WECT) at comfortable speed to assess VO_{2walk} and walking speed, followed by a maximal incremental exercise test to assess VO_{2peak}. In addition, daily physical activity during waking hours was measured for 7 consecutive days.

Physical strain of walking. The physical strain of walking (in % of VO_{2peak}) was calculated as the ratio of VO_{2walk} (measured during the WECT) and VO_{2peak} (measured during the maximal incremental exercise test), multiplied by 100%.

Oxygen consumption of walking (VO_{2walk}). After an 8-minute rest period, participants performed a 6-minute WECT at selfselected, comfortable speed on a 30-meter indoor track. The use of assistive devices (e.g., cane, stick, walker) during the test was allowed. Throughout the test, breath-by-breath oxygen consumption was measured with a portable gas analysis system (K5 Gas Analysis System, Cosmed, Rome, Italy). Before every test, the gas analysis system was calibrated according to the manual instructions. We visually determined a steady state period of at least 60 seconds between the third and sixth minute of the test and calculated the mean oxygen consumption (VO_{2walk} , in ml/kg/ min) over this steady-state period. Walking speed (in m/s) was calculated as the total distance walked divided by 360 seconds. Previously, the WECT with simultaneous breath-by-breath gas analysis has been shown to be valid for measuring VO_{2walk} in people with NMD (16).

*Peak oxygen uptake (VO*_{2peak}). After screening for contraindications by a physician, a maximal incremental exercise test on a bicycle ergometer (Lode Excalibur, Lode BV, Groningen, The Netherlands) was executed according to international guidelines concerning standardization, and supervised by a trained researcher (28). After 3 minutes of rest, the test started with 3 minutes unloaded cycling, followed by 5–20 Watt/minute continuous workload increments, depending on the participants' physical fitness level. During the test, oxygen uptake was continuously measured using MasterScreen CPX (CareFusion, Hoechberg, Germany) (30). The test was stopped if 1 of the following criteria was met: oxygen uptake plateau reached, pedal frequency dropping below 50 revolutions per minute, or the participant met the ACSM stop criteria for exercise testing (28). VO_{2peak} (in ml/kg/min) was determined as the highest 30 s moving average. Daily physical activity. Daily physical activity was determined by step count and time spent in MVPA. Participants were provided with an accelerometer (ActiGraph GT3X+ accelerometer, Health One Technology, Fort Walton Beach, FL, USA) and a heart rate belt (Polar H10, Polar Electro, Kempele, Finland). They were asked to wear the accelerometer on the left side of their waist and the heart rate belt just below their sternum for all waking hours (except for bathing and swimming) for 7 consecutive days (27). A demonstration was given on proper positioning of both devices and participants were provided with written instructions on placement and wear time of the device. The accelerometer and heart rate belt were connected and the data were analysed with the ActiLife software (version 6.13.4, Health One Technology, Fort Walton Beach, FL, USA). Physical activity data were used if at least 4 days of 480 minutes or more were measured. From the measured data we calculated the mean daily step count. Intensity zones were individually calculated based on percentages of the heart rate reserve (HRR), defined as the peak heart rate during the maximal incremental exercise test minus the resting heart rate. Intensity zones were categorized as low (< 40% HRR), moderate (40-60% HRR), and vigorous (> 60% HRR) (22). For this study, we summed the daily time spent in the moderate and vigorous activity zones as time spent in MVPA.

Statistical analysis

Descriptive data were expressed as mean and standard deviation (SD) in the case of normally distributed data, or as median and interquartile range (IQR) if data were not normally distributed, based on visual inspection. To determine whether the physical strain of walking was increased in our sample of subjects with NMD, we compared the outcome with previously published reference data of healthy subjects of similar age (18, 19).

Relationships between daily step count and daily time spent in MVPA (dependent variables) with physical strain (independent variable) were assessed with linear regression analyses (2 separate models). If the dependent variables were not normally distributed, they were transformed using the natural-log function to approximate a normal distribution. Regression analyses were checked for confounding effects of gender, age, body mass index (BMI), MRC sum score of lower extremities, and comfortable walking speed. Each potential confounder was entered separately into the uncorrected model and the strongest confounder that changed the β by at least 10% was added to the model. This procedure was repeated until there were no more confounders to add to the corrected model. To explore effect modification, we added the interaction of muscle strength (MRC sum score), which we considered to be the most plausible effect modifier, to the corrected models, and reported whether it was an effect modifier or not. All analyses were performed using SPSS (version 26.0, IBM Corp, Armonk, NY, USA). A p-value of < 0.05 was considered statistically significant.

Table I. Characteristics of subjects

Sex (male/female), n	24/37
Age (years), median (IQR range)	63.0 (50.5-67.0)
Height (cm), mean (SD)	171.7 (10.5)
Weight (kg), mean (SD)	75.8 (14.2)
BMI, mean (SD)	25.8 (4.9)
MRC sum lower extremities (range: 0-80), median (IQR range)	73.3 (63.2–76.8)
Walking aid (yes/no), n	45/16
Diagnosis (PPS/CMT/other), n	14/31/16

BMI: body mass index, CMT: Charcot-Marie-Tooth disease, MRC: Medical Research Council scale, PPS: post-polio syndrome.

Table II. Physical strain of walking and daily activity outcomes

Physical strain of walking (%), mean (SD)	72.6 (20.3)	
VO _{2walk} (ml/min/kg), mean (SD)	14.9 (2.9)	
VO _{2peak} (ml/min/kg), mean (SD)	21.7 (7.8)	
Walking speed (m/s), mean (SD)	1.0 (0.2)	
Daily step count, median (IQR range)	4563 (3567-6664)	
Time spent in MVPA (min per day), median (IQR range)	55 (25-109)	
ml: millilitre, min: minute, kg: kilogram, m: meter, s: second, h: hour, MVPA: moderate and vigorous physical activity zones.		

RESULTS

Of the 91 subjects included in the I'M FINE study, 61 had complete and valid baseline data for all variables of interest. Incomplete or invalid data were due to missing or incomplete daily activity data (n=23), invalid VO_{2walk} determination (n=5), or non-ambulatory participants (n=2). There were no significant differences in subject characteristics between the groups included (n=61) and excluded (n=30) for the analyses.

Subjects (37 females) were diagnosed with Charcot-Marie-Tooth disease (n=31), post-polio syndrome (n=14), or other NMD (n=16) (Table II). Participants with "other NMD" were diagnosed with congenital myopathy (n=5), limb-girdle muscular dystrophy (n=4), myotonic dystrophy (n=3), inclusion body myositis (n=2), oculopharyngeal muscular dystrophy (n=1), and chronic idiopathic axonal polyneuropathy (n=1). We observed MRC sum scores below 80 in 54 participants (89%). The median (IQR) MRC sum score was 73.3 (63.2–76.8).

The mean (SD) physical strain of walking was 73 (20)%, and 34–49% in control groups of able-bodied individuals of similar age (18, 19) (Table II). The median (IQR) daily step count was 4,563 (3,567–6,664) steps, and 55 (25–109) minutes spent daily in MVPA.

Daily step count and time spent in MVPA were both not normally distributed and these variables were therefore transformed using the natural-log function. We found a significant association between log daily step count and physical strain of walking, in both the corrected and uncorrected regression analyses (Table III). The corrected regression analysis showed an inverse relationship (β =-0.47) between log daily step count and physical strain of walking. No significant asso-

Table III. Linear regression analyses with log daily step count and log time spent in MVPA as dependent variables and physical strain of walking as independent variable

Factor	Regression coefficient (β)	95%-CI	<i>p</i> -value
Log daily step count and physical strain (uncorrected)	-0.58	-0.79; -0.37	< 0.001
Log daily step count and physical strain (corrected) ^a	-0.47	-0.67; -0.27	< 0.001
Log time spent in MVPA and physical strain (uncorrected)	0.21	-0.05; 0.46	0.11
Log time spent in MVPA and physical strain (corrected) ^b	0.12	-0.15; 0.40	0.37

^aCorrected for age, walking speed. ^bCorrected for gender, age, body mass index. CI: confidence interval, MVPA: moderate to vigorous physical activity. ciation between log daily time spent in MVPA and physical strain of walking was found (Table III). There were no significant interactions between lower extremity MRC sum score and the independent variables in the corrected analyses.

DISCUSSION

Our study in ambulatory people with different NMDs showed that physical strain of comfortable walking was severely increased, indicating that walking is highly demanding. Log daily step count was significantly inversely related to physical strain of walking, while log time spent in MVPA was not related to physical strain of walking.

The mean physical strain of walking of 73% we found is comparable to the physical strain of walking of 65% reported in polio survivors, and much higher than the reportedly 34–49% in able-bodied individuals of similar age (18, 19). It is also considerably higher than the physical strain of 50–52% found in other patient groups, including children and adults with cerebral palsy (10, 31), and adults post-stroke (18). The results from our study and previous studies imply that walking for people with NMD is highly demanding, and this may negatively influence health, daily life activities, and quality of life (32, 33).

A low VO_{2peak} rather than a high VO_{2walk} seemed to be the main contributor to the high physical strain of walking in our sample. Comparable to previous studies in NMD (11), we found a VO_{2neak} of 21.7 ml/ min/kg, which is on average 30–44% lower than the VO_{2peak} reported in healthy subjects of similar age (34), possibly caused by the reduced muscle mass due to the progressing muscle weakness in NMD (17). The VO_{2walk} in our sample was not increased and comparable to values found in healthy subjects of similar age (18, 19). This likely resulted from the 20-26%reduction in walking speed in our subjects with NMD, as compared with healthy subjects (18, 19). The latter finding is in line with the energy minimization theory (35) and earlier research in NMD (36), showing that people with NMD walk slower to prevent increases in the oxygen consumption of walking, so that it can be sustained for longer periods in daily life. To lower the high physical strain of walking in people with NMD, neuromuscular rehabilitation could focus on increasing the VO_{2neak} by means of aerobic exercise. Based on a recent systematic review on aerobic capacity in NMD (11), this may be especially beneficial in people with lower baseline VO_{2peak} levels compared with those with higher baseline VO_{2peak} levels. Another approach to reduce the physical strain of walking could be to the decrease the VO_{2walk} in those people with an elevated VO_{2walk} and/or to increase walking speed, without affecting the VO_{2walk} , for example by prescribing or optimizing lower extremity orthoses (37).

The impact of a high physical strain of walking on daily activity in our subjects with NMD was highlighted by the inverse relation with the log daily step count, i.e., people with a high physical strain of walking took fewer steps per day. This is in line with the inverse relation between physical strain of walking and total daily walking time reported in adults with cerebral palsy (10). However, no relationship was found between physical strain of walking and the log daily time spent in MVPA in our study. This may indicate that, despite a reduction in daily step activity, strenuous daily activities may still be performed. An earlier study on Charcot-Marie-Tooth disease (25) also found a lower daily step count, but no differences in time spent in MVPA, as compared with healthy subjects. As suggested by the authors, the difference between the 2 daily activity outcomes could be due to people breaking up sedentary periods with more frequent transitions into standing, without accumulating a larger number of steps. In line with their recommendation, more detailed exploration with accelerometers that can distinguish between different activities, such as lying, sitting, and standing, would be required. Furthermore, they suggested an overestimation of the time spent in MVPA as the activity monitor used was not validated in the NMD population and did not correct for potential reduced energy expenditure due to reduced muscle mass. However, this suggestion seems unlikely as we found similar results while using individually determined time spent in MVPA, based on heart rate measurements. Alternatively, in line with Horemans et al. (33), it should be realized that intensity of daily-life activities is not only determined by walking ability but also by other factors, such as social behaviour, personal lifestyle, working conditions, and living circumstances. Moreover, certain tasks have to be done in daily life, despite high intensity or strain.

Limitations

First, we used baseline data of an intervention study for people with NMD who were motivated to increase their aerobic capacity. Moreover, all participants were ambulant, limiting generalizability to other, for instance non-ambulant, people with NMD. It would be interesting for future research to investigate the relationship between the physical strain of wheelchair use and its relationship with daily activity. Second, while we were interested in the physical strain of walking, the VO_{2peak} was determined during cycling and not during walking. As the measured VO_{2peak} during walking is generally higher than the VO_{2peak} during cycling (38), due to a larger exercising muscle mass during walking,

JRN

this may have led to an overestimation of the physical strain of walking. Finally, although the accelerometer we used to determine steps has often been used in earlier studies assessing daily-life physical activity with proven good validity for assessing daily step count in different elderly populations, its validity to assess step count at lower walking speeds (< 0.9 m/s) is low (39). Some of our subjects had a walking speed lower than 0.9 m/s, which could have influenced the investigated relation between the physical strain of walking and daily steps taken.

Conclusion

Our study showed that the physical strain of comfortable walking was severely increased. On average the physical strain was 73%, indicating that walking is highly demanding in people with NMD. The number of daily steps taken was inversely related to the physical strain of comfortable walking, while intensity of daily activity was not related. Thus, people with increased physical strain took fewer steps, but the absence of a relationship with moderate and vigorous intensive activities may indicate that, despite a reduction in daily step activity, strenuous daily activities may still be performed. Reducing the physical strain of walking by increasing the VO_{2peak} with aerobic exercise or by reducing the VO_{2walk} by means of lower limb orthoses should be further investigated as potential clinical directions to improve daily activity in people with NMD, and studies with a longitudinal design should focus on the causal relation between physical strain of walking and daily activity.

ACKNOWLEDGEMENTS

Funding: This work was supported by the Prinses Beatrix Spierfonds (grant number W.OK17–3), received by EV, MB and FN.

Trial details: Dutch Trial register, NL7344.

The authors have no conflicts of interest to declare.

REFERENCES

- Booth FW, Roberts CK, Laye MJ. Lack of exercise is a major cause of chronic diseases. Compr Physiol 2012; 2: 1143–1211. DOI: 10.1002/cphy.c110025
- Phillips M, Flemming N, Tsintzas K. An exploratory study of physical activity and perceived barriers to exercise in ambulant people with neuromuscular disease compared with unaffected controls. Clin Rehabil 2009; 23: 746–755. DOI: 10.1177/0269215509334838
- McDonald CM. Physical activity, health impairments, and disability in neuromuscular disease. Am J Phys Med Rehabil 2002; 81: S108–120. DOI: 10.1097/00002060-200211001-00012
- Winberg C, Flansbjer UB, Rimmer JH, Lexell J. Relationship between physical activity, knee muscle strength, and gait performance in persons with late effects of polio. PM

R 2015; 7: 236-244. DOI: 10.1016/j.pmrj.2014.09.005

- Féasson L, Camdessanché JP, El Mandhi L, Calmels P, Millet GY. Fatigue and neuromuscular diseases. Ann Readapt Med Phys 2006; 49: 289–300. DOI: 10.1016/j. annrmp.2006.04.015
- Jensen MP, Abresch RT, Carter GT, McDonald CM. Chronic pain in persons with neuromuscular disease. Arch Phys Med Rehabil 2005; 86: 1155–1163. DOI: 10.1016/j. apmr.2004.11.028
- Andries A, Van Walsem MR, Ørstavik K, Frich JC. Functional ability and physical activity in hereditary neuromuscular diseases. J Neuromuscul Dis 2022; 9: 437–446. DOI: 10.3233/JND-210677
- Winberg C, Brogårdh C, Flansbjer UB, Carlsson G, Rimmer J, Lexell J. Physical activity and the association with selfreported impairments, walking limitations, fear of falling, and incidence of falls in persons with late effects of polio. J Aging Phys Act 2015; 23: 425–432. DOI: 10.1123/ japa.2014-0163
- Anens E, Emtner M, Hellström K. Exploratory study of physical activity in persons with Charcot-Marie-Tooth disease. Arch Phys Med Rehabil 2015; 96: 260–268. DOI: 10.1016/j.apmr.2014.09.013
- Slaman J, Bussmann J, van der Slot WM, Stam HJ, Roebroeck ME, van den Berg-Emons RJ. Transition and Lifespan Research Group South West Netherlands. Physical strain of walking relates to activity level in adults with cerebral palsy. Arch Phys Med Rehabil 2013; 94: 896–901. DOI: 10.1016/j.apmr.2012.11.005
- Oorschot S, Brehm MA, Daams J, Nollet F, Voorn EL. Efficacy of aerobic exercise on aerobic capacity in slowly progressive neuromuscular diseases: a systematic review and meta-analysis. Ann Phys Rehabil Med 2023; 66: 101637. DOI: 10.1016/j.rehab.2022.101637
- Wiesinger GF, Quittan M, Nuhr M, Volc-Platzer B, Ebenbichler G, Zehetgruber M, et al. Aerobic capacity in adult dermatomyositis/polymyositis patients and healthy controls. Arch Phys Med Rehabil 2000; 81: 1–5. DOI: 10.1016/ s0003-9993(00)90212-0
- Brehm MA, Nollet F, Harlaar J. Energy demands of walking in persons with postpoliomyelitis syndrome: relationship with muscle strength and reproducibility. Arch Phys Med Rehabil 2006; 87: 136–140. DOI: 10.1016/j. apmr.2005.08.123
- Gait Analysis: Normal and pathological function. J Sports Sci Med 2010; 9: 353.
- Ploeger HE, Bus SA, Nollet F, Brehm MA. Gait patterns in association with underlying impairments in polio survivors with calf muscle weakness. Gait Posture 2017; 58: 146–153. DOI: 10.1016/j.gaitpost.2017.07.107
- Brehm MA, Verduijn S, Bon J, Bredt N, Nollet F. Comparison of two 6-minute walk tests to assess walking capacity in polio survivors. J Rehabil Med 2017; 49: 732–737. DOI: 10.2340/16501977-2264
- 17. Nollet F, Beelen A, Sargeant AJ, de Visser M, Lankhorst GJ, de Jong BA. Submaximal exercise capacity and maximal power output in polio subjects. Arch Phys Med Rehabil 2001; 82: 1678–1685. DOI: 10.1053/ apmr.2001.27390
- Blokland I, Gravesteijn A, Busse M, Groot F, van Bennekom C, van Dieen J, et al. The relationship between relative aerobic load, energy cost, and speed of walking in individuals post-stroke. Gait Posture 2021; 89: 193–199. DOI: 10.1016/j.gaitpost.2021.07.012
- Wezenberg D, van der Woude LH, Faber WX, de Haan A, Houdijk H. Relation between aerobic capacity and walking ability in older adults with a lower-limb amputation. Arch Phys Med Rehabil 2013; 94: 1714–1720. DOI: 10.1016/j. apmr.2013.02.016
- Apabhai S, Gorman GS, Sutton L, Elson JL, Plötz T, Turnbull DM, et al. Habitual physical activity in mitochondrial disease. PLoS One 2011; 6: e22294. DOI: 10.1371/journal. pone.0022294

- Bull FC, Al-Ansari SS, Biddle S, Borodulin K, Buman MP, Cardon G, et al. World Health Organization 2020 guidelines on physical activity and sedentary behaviour. Br J Sports Med 2020; 54: 1451–1462. DOI: 10.1136/ bjsports-2020-102955
- MacIntosh BR, Murias JM, Keir DA, Weir JM. What is moderate to vigorous exercise intensity? Front Physiol 2021; 12: 682233. DOI: 10.3389/fphys.2021.682233
- Winberg C, Flansbjer UB, Carlsson G, Rimmer J, Lexell J. Physical activity in persons with late effects of polio: a descriptive study. Disabil Health J 2014; 7: 302–308. DOI: 10.1016/j.dhjo.2014.02.003
- Andersen LK, Vissing J. Habitual physical activity in patients with myasthenia gravis assessed by accelerometry and questionnaire. J Neuromuscul Dis 2022; 9: 161–169. DOI: 10.3233/JND-210693
- Ramdharry GM, Pollard AJ, Grant R, Dewar EL, Laurá M, Moore SA, et al. A study of physical activity comparing people with Charcot-Marie-Tooth disease to normal control subjects. Disabil Rehabil 2017; 39: 1753–1758. DOI: 10.1080/09638288.2016.1211180
- Knak KL, Sheikh AM, Witting N, Vissing J. Physical activity in myotonic dystrophy type 1. J Neurol 2020; 267: 1679–1686. DOI: 10.1007/s00415-020-09758-8
- Oorschot S, Brehm MA, van Groenestijn AC, Koopman FS, Verhamme C, Eftimov F, et al. Efficacy of a physical activity programme combining individualized aerobic exercise and coaching to improve physical fitness in neuromuscular diseases (I'M FINE): study protocol of a randomized controlled trial. BMC Neurol 2020; 20: 184. DOI: 10.1186/ s12883-020-01725-0
- Ferguson B. ACSM's Guidelines for exercise testing and prescription. 9th ed. J Can Chiropr Assoc 2014; 58: 328.
- 29. Compston A. Aids to the investigation of peripheral nerve injuries. Medical Research Council: Nerve Injuries Research Committee. His Majesty's Stationery Office: 1942; pp. 48 (iii) and 74 figures and 7 diagrams; with aids to the examination of the peripheral nervous system. By Michael O'Brien for the Guarantors of Brain. Saunders Elsevier: 2010; pp. [8] 64 and 94 Figures. Brain 2010; 133: 2838–2844. DOI: 10.1093/brain/awq270
- 30. Rietjens GJ, Kuipers H, Kester AD, Keizer HA. Validation of

a computerized metabolic measurement system (Oxycon-Pro) during low and high intensity exercise. Int J Sports Med 2001; 22: 291–294. DOI: 10.1055/s-2001-14342

- Dallmeijer AJ, Brehm MA. Physical strain of comfortable walking in children with mild cerebral palsy. Disabil Rehabil 2011; 33: 1351–1357. DOI: 10.3109/09638288.2010.531374
- Brehm MA, Ploeger HE, Nollet F. Self-reported functional ambulation is related to physical mobility status in polio survivors: a cross-sectional observational study. Ann Phys Rehabil Med 2021; 64: 101428. DOI: 10.1016/j. rehab.2020.06.007
- Horemans HL, Bussmann JB, Beelen A, Stam HJ, Nollet F. Walking in postpoliomyelitis syndrome: the relationships between time-scored tests, walking in daily life and perceived mobility problems. J Rehabil Med 2005; 37: 142–146. DOI: 10.1080/16501970410021526
- 34. van der Steeg GE, Takken T. Reference values for maximum oxygen uptake relative to body mass in Dutch/Flemish subjects aged 6–65 years: the LowLands Fitness Registry. Eur J Appl Physiol 2021; 121: 1189–1196. DOI: 10.1007/ s00421-021-04596-6
- Waters RL, Lunsford BR, Perry J, Byrd R. Energy-speed relationship of walking: standard tables. J Orthop Res 1988; 6: 215–222. DOI: 10.1002/jor.1100060208
- Menotti F, Felici F, Damiani A, Mangiola F, Vannicelli R, Macaluso A. Charcot-Marie-Tooth 1A patients with low level of impairment have a higher energy cost of walking than healthy individuals. Neuromuscul Disord 2011; 21: 52–57. DOI: 10.1016/j.nmd.2010.09.008
- Waterval NFJ, Brehm MA, Harlaar J, Nollet F. Description of orthotic properties and effect evaluation of ankle–foot orthoses in non-spastic calf muscle weakness. J Rehabil Med 202018; 52: jrm00026. DOI: 10.2340/16501977–2642
- 38. Shephard RJ. Tests of maximum oxygen intake: a critical review. Sports Med 1984; 1: 99–124. DOI: 10.2165/00007256-198401020-00002
- Ngueleu AM, Barthod C, Best KL, Routhier F, Otis M, Batcho CS. Criterion validity of ActiGraph monitoring devices for step counting and distance measurement in adults and older adults: a systematic review. J Neuroeng Rehabil 2022; 19: 112. DOI: 10.1186/s12984-022-01085-5